Author's response to reviews

Title: The effects of multi-domain versus single-domain cognitive training in non-demented older people: a randomized controlled trial

Authors:

Yan Cheng (yancheng_research@hotmail.com)
Wenyuan Wu (wenyuanwu_tj@hotmail.com)
Wei Feng (ffww06@163.com)
Jiaqi Wang (jiamian316@163.com)
You Chen (huaganyan@163.com)
Yuan Shen (shenyuanmm@yahoo.com.cn)
Qingwei Li (lianocd@yahoo.com.cn)
Xu Zhang (zzz1082@163.com)
Chunbo Li (chunbo_li@yahoo.com)

Version: 5 Date: 22 February 2012

Author's response to reviews: see over
Dear Dr. Tree-Booker,

Thank you very much for the suggestions (MS: 5651524986181287). We have added the abbreviations to the table legends as suggested.

Sincerely yours,

Wenyuan Wu